Author's response to reviews

Title: Overexpression p21WAF1/CIP1 in suppressing retinal pigment epithelial cells and progression of proliferative vitreoretinopathy via inhibition CDK2 and cyclin E

Authors:

Ying Wang (wangying_1026@yahoo.com)
Zhigang Yuan (yzq0158@163.com)
Caiyun You (youcaiyun@126.com)
Jindong Han (djindonghan@126.com)
Haiyan Li (lhyeye@163.com)
Zhuhong Zhang (zhangzhuhonghao@163.com)
Hua Yan (phuayan2000@163.com)

Version: 7 Date: 4 November 2014

Author's response to reviews: see over
Dear Editor,

Thank you editor for constructive comments and suggestions to this manuscript (MS: 3903436101358187). The correction was made according to your comments. We’ll try our best to answer or explain all questions you and reviewers have presented. Below are our explanations on comments reviewer(s) presented.

Sincerely,
Ying Wang

Editorial Revisions:
Thank you for your cover letter and revised manuscript, however please include the two approval/permit numbers from the medical ethics committee and animal ethics committee within your main manuscript.

Reply: The permit numbers from the animal ethics committee is TMUaMEC2013008. The approval numbers from the medical ethics committee is 201315. I have added them in my manuscript. I will upload the documents of approval if necessary.

I hope that the correction will meet with approval. Once again, thank you very much for your comments and suggestions.